Author’s response to reviews

Title: A new scoring system to stratify risk in unstable angina.

Authors:

Dr Alfredo C Piombo (apiombo@intramed.net.ar)
Juan A Gagliardi (jag@cardioweb.net.ar)
Javier Guetta (guetta2000@yahoo.com.ar)
Juan J Fuselli (jfuselli@cemic.com.ar)
Simon Salzberg (simon.salzberg@roche.com.ar)
Enrique Fairman (efairman@intramed.net.ar)
Carlos A Bertolasi (hauc@intramed.net.ar)

Version: 3 Date: 28 Mar 2003

All the compulsory revisions and many of the discretionary ones suggested by the reviewers have been incorporated to the text or tables.
We tried to please the three reviewers' suggestions, although in some cases it may be rather difficult to do so because their opinions are not necessarily identical.